Author’s response to reviews

Title: A clinical prediction rule to identify difficult intubation in children with Robin sequence requiring mandibular distraction osteogenesis based on craniofacial CT measures

Authors:
Zhe Mao (39107832@qq.com)
Na Zhang (na999666@163.com)
Yingqiu Cui (gzhtwang@163.com)

Version: 1 Date: 28 Jun 2019

Author’s response to reviews:

1. Overlap: Thank you for your comment. This issue had been resolved, and the sections of text in question have been reformulated.

2. Authors’ contributions: I had represented authors’ names using their full initials. All authors have read and approved the manuscript.

3. Formatting: “Information and methods” had been renamed by “Methods”.

4. Clean Manuscript: A clean version of our manuscript had been submitted without track changes or highlighting.